Author's response to reviews

Title: When care situations evoke difficult emotions in nursing staff members: an ethnographic study in two Norwegian nursing homes

Authors:

Anne Marie A.M.S. Sandvoll (annemsa@hisf.no)
Solveig S.H. Hauge (solveig.hauge@hit.no)
Kjell K.K Kristoffersen (kjell.kristoffersen@uia.no)
Ellen Karine E.K.G. Grov (Ellen.karine.grov@hisf.no)

Version: 6 Date: 8 June 2015

Author's response to reviews: see over
Dear Miss Nawsheen Boodhun

BMC Nursing

Førde, June 8th, 2015

Thank you for the editorial comments to our revised manuscript “When care situations evoke difficult emotions in nursing staff members: an ethnographic study in two Norwegian nursing homes”.

We have considered the comments and we are pleased to resubmit a revised article with the following changes (marked in red), made point-by-point according to the editor’s suggestions.

Editorial requirements:

1. Please include the title page at the beginning of the main manuscript file and remove it from the additional files. Also, include the e-mails of all authors on the title page. The title page has been included with e-mail of all authors.
2. Kindly include the name of the Regions Medical Ethics. The committee in charge of approval for this study: Regional Ethics Committee (REK) West, University of Bergen. The information is added.
3. Clarify if consent was informed. Done as suggested by the reviewer.
4. In the Acknowledgment, please clarify who provided funding. This information has been added.
5. Abbreviations section. We have only one abbreviation, which is explained in the text. Since it is just one abbreviation, putting it into a separate section seems unnecessary.